Author's response to reviews

Title: Predictors of Metabolic Monitoring among Schizophrenia Patients Initiated on Second Generation Antipsychotics in the Veterans Health Administration

Authors:

Lizheng Shi (lshi1@tulane.edu)
Haya Ascher-Svanum (Ascher-Svanum_Haya@lilly.com)
Yi-Ju Chiang (chiang_642@hotmail.com)
Yingnan Zhao (yzhao3@tulane.edu)
Vivian Fonseca (vfonseca@tulane.edu)
Daniel Winstead (winstead@tulane.edu)

Version: 2 Date: 30 July 2009

Author's response to reviews: see over
July 10, 2009

To whom it may concern:

We are electronically re-submitting our manuscript titled, “Predictors of Metabolic Monitoring among Schizophrenia Patients Initiated on Atypical Antipsychotics in the Veterans Health Administration.”

We have addressed the comments from two knowledgeable reviewers (see response letter). Particularly, we have re-run the logistical regression model and use three different metabolic syndrome ascertaining methods. We have included both track-changed versions and clean versions for manuscript text and tables.

We would appreciate your review and consideration of this manuscript and look forward to hearing from you regarding possible publication. Please address correspondence to Lzheng Shi, PhD, MsPharm; Assistant Professor; Department of Health Systems Management; Tulane University; 1440 Canal Street, Suite 1900; New Orleans, LA 70112; Tel: (504)-988-6548; Fax: (504)-988-3783; Email: lshi1@tulane.edu.

Sincerely,

Lzheng Shi, PhD, MsPharm
Tulane University